
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log
-----------------------------------------------------------------------------
        Fri Aug 16 16:12:06 UTC 2013
Year: 13  Doy: 228
Observer: FakeObserver
Fri Aug 16 06:38:53 HST 2013: PSPT Start Patrol
****EVENT COMMENT BY berkey**** : Fri Aug 16 07:55:43 HST 2013
A non event happened.
___end___
WEATHER COMMENT: berkey: Fri Aug 16 08:04:52 HST 2013
coronal skies, temp 50, no wind
___end___
CoMP COMMENT BY berkey: Fri Aug 16 08:08:53 HST 2013
No CoMP obs today, due to lack of guider and warm camera. 
___end___
PSPT COMMENT BY berkey: Fri Aug 16 08:09:15 HST 2013
Pspt looks like it is working.
___end___
**CoMP PROBLEM COMMENT BY berkey **: Fri Aug 16 08:09:26 HST 2013
All the CoMP electrics are off.  Observing canceled today.
___end___
**K-Cor PROBLEM COMMENT BY berkey **: Fri Aug 16 08:10:10 HST 2013
K-Cor is in boxes no observing at either spar today.
___end___
K-Cor COMMENT BY berkey: Fri Aug 16 08:10:47 HST 2013
Rack space for kcor at mlso is clear and avaiable
___end___
GONG COMMENT BY berkey: Fri Aug 16 08:11:24 HST 2013
Gong looks fine
___end___
**GONG PROBLEM COMMENT BY berkey** : Fri Aug 16 08:11:32 HST 2013
No problem here.
___end___
**PSPT PROBLEM COMMENT BY berkey** : Fri Aug 16 08:12:05 HST 2013
Pspt looks fine right now, but it MAY crash at some point today.
___end___
GENERAL OBSERVATORY COMMENT BY berkey: Fri Aug 16 08:12:39 HST 2013
I found a home for the old 19" half racks in NOAA storage.  They are on the mountain and avaiable when needed but wont tak
e up MLSO realestate.  (Note there are 4 of these weird 1/2 rack sections that used to make up the nahenahe/kii observatio
n station. Not the typical full height racks.  When needed (say for FPI) we can go down to the storage room and pick them 
up.
___end___
**GENERAL PROBLEM COMMENT BY berkey** : Fri Aug 16 08:14:56 HST 2013
We aren’t doing any spar observations due to lack of guider.  Too bad with such nice skies today.
___end___
Fri Aug 16 08:48:51 HST 2013: PSPT Start Patrol
Possible CME in Progress berkey: Fri Aug 16 11:48:46 HST 2013



*****CME COMMENT*****
A Cme was not seen, we have no cornal insturments today.  This is a tese of a command.
___end___
  
   You can check out the image data at: 
        http://mlso.hao.ucar.edu 
Possible CME in Progress berkey: Fri Aug 16 11:49:56 HST 2013
*****CME COMMENT*****
A Cme was not seen, we have no cornal insturments today.  This is a tese of a command.
___end___
  
   You can check out the image data at: 
        http://mlso.hao.ucar.edu 
Possible CME in Progress berkey: Fri Aug 16 11:52:35 HST 2013
*****CME COMMENT*****
testing
___end___
  
   You can check out the image data at: 
        http://mlso.hao.ucar.edu 
Possible CME in Progress berkey: Fri Aug 16 11:53:08 HST 2013
*****CME COMMENT*****
One more test
___end___
  
   You can check out the image data at: 
        http://mlso.hao.ucar.edu 
CoMP COMMENT BY berkey: Fri Aug 16 11:54:47 HST 2013
asdf
___end___
CoMP COMMENT BY berkey: Fri Aug 16 11:55:03 HST 2013
test
___end___
**CoMP PROBLEM COMMENT BY berkey **: Fri Aug 16 11:55:40 HST 2013
test
___end___
**GONG PROBLEM COMMENT BY berkey** : Fri Aug 16 11:55:55 HST 2013
test
___end___
GONG COMMENT BY berkey: Fri Aug 16 11:56:14 HST 2013
Test
___end___
**GENERAL PROBLEM COMMENT BY berkey** : Fri Aug 16 11:56:54 HST 2013
Test
___end___
**PSPT PROBLEM COMMENT BY berkey** : Fri Aug 16 11:57:28 HST 2013
test
___end___



PSPT COMMENT BY berkey: Fri Aug 16 11:57:47 HST 2013
Testing
___end___


